Author's response to reviews

Title: Efficacy of vitamin C as an adjunct to fluoxetine therapy in pediatric major depressive disorder: a randomized, double-blind, placebo-controlled pilot study

Authors:

Mostafa Amr (mamr@gnif.org)
Ahmed El-Mogy (aelmogy@gnif.org)
Tarek Shams (tshams@gnif.org)
Karen Vieira (kvieira@gnif.org)
Shaheen E Lakhan (slakhan@gnif.org)

Version: 4 Date: 15 February 2013

Author's response to reviews:

Dear Editor,

We appreciate the final minor revision request made by reviewer 1 and that reviewer 2 found no further edits needed. Specifically, we have corrected the error in the methods section to now properly read that two subjects from the Vitamin C group and one from the placebo group were removed from the study due to non-compliance (page 5). This is reflected in the results section too (page 7).

We hope that the paper is acceptable for publication with this final edit. Thank you.

Sincerely,
Shaheen E Lakhan